STRATOSPHERIC DYNAMICS

Peter Haynes

Department of Applied Mathematics and Theoretical Physics, University of Cambridge, Cambridge CB3 0WA, United Kingdom; email: phh@damtp.cam.ac.uk

Key Words geophysical fluid dynamics, vortex dynamics, climate, weather, atmosphere

■ Abstract The large-scale dynamics of the extratropical stratosphere are reviewed. The role of Rossby waves and vortex dynamics in shaping the winter stratospheric circulation and the dynamics of the longitudinal mean flow are first discussed separately. The important effects of two-way interaction between waves and mean flow are then described, with emphasis on how mechanisms discovered in simple models can be followed through to models that are closer to the real stratosphere. A final topic is the possible effect of the stratosphere on the troposphere, with emphasis on dynamical mechanisms for such an effect.

1. INTRODUCTION

The atmosphere is conventionally subdivided in the vertical into regions according to the vertical temperature gradient, with temperature decreasing upward in the troposphere (0-10 km in global average), then increasing upwards in the stratosphere (10–50 km), then decreasing upward in the mesosphere (50–80 km), and then increasing upward again in the lower thermosphere (above 80 km). In the textbook Middle Atmosphere Dynamics (Andrews et al. 1987), the authors use the term "middle atmosphere" to describe the last three regions (up to about 100 km). The middle atmosphere can be distinguished from the troposphere below by the fact that it is remote from the Earth's surface and the dynamical effect of latent heating due to phase changes of water can be neglected (whereas in the troposphere it is extremely important). The upper limit corresponds roughly to the homopause, above which molecular processes rather than bulk fluid processes dominate transport, allowing differential gravitational settling of different chemical species according to molecular weight, and also to the level above which increased ionization implies a strong role for electromagnetic forces in the dynamics. The middle atmosphere can be approximated as well-mixed on small scales and electrically neutral.

This review focuses primarily on the lowest part of the middle atmosphere, the stratosphere, which has the potential significantly to affect conditions at the surface, as follows:

- (a) Stratospheric ozone: Changes in stratospheric ozone imply changes in surface ultraviolet irradiance and also a changed supply of ozone to the troposphere. The latter may have a significant effect on surface ozone concentrations (Zeng & Pyle 2003).
- (b) Radiative balance of the troposphere: The lower stratosphere plays an important role in the radiative balance of the troposphere, and changes in greenhouse gases such as ozone and water vapor in the lower stratosphere directly affect surface temperatures.
- (c) Weather and climate: Whereas in weather forecasting and tropospheric climate modelling it has been customary to de-emphasize the role of the stratosphere, there is now evidence of downward dynamical links between the stratosphere and troposphere. These links may determine, for example, the effect on surface weather and climate of stratospheric aerosol changes due to volcanic eruptions and may imply a strong role for the stratosphere in determining future changes in the tropospheric climate due to increases in carbon dioxide and other greenhouse gases.
- (d) Solar variability: Vertical coupling of dynamics between stratosphere and troposphere may provide an explanation for apparent signals of solar variability (whose direct physical and chemical effects on the mesosphere and upper stratosphere are significant and relatively well-understood) in tropospheric weather and climate.

Study of the stratosphere over the last two decades has been stimulated primarily by (a, above) the need to account for observed changes in stratospheric ozone (most famously the Antarctic ozone hole) and to determine the relation to anthropogenic chemical emissions, particularly emissions of long-lived halogen compounds such as halocarbons. The rate of emission of such compounds has now decreased, following the Montreal Protocol, and attention is now focused on the "recovery" of the ozone layer, as halogen concentrations in the stratosphere decrease. There is still concern that some of the shorter-lived replacements may cause significant ozone destruction (WMO 2003), but attention is shifting to the possible impact of future climate change on ozone concentrations. Meanwhile, the growing evidence that the stratosphere plays a wider role in the chemical-climate system (b, c, d, above) has stimulated new interest in this region.

Andrews et al. (1987) reviewed basic principles of the dynamics¹ of the middle atmosphere that remain relevant today. However, there have been many important developments since, including the period of intense research into stratospheric

¹The convention in atmospheric science is to divide physical processes into dynamics on the one hand, meaning the part described by the momentum and thermodynamic equations (i.e., fluid dynamics), and 'physics' on the other, meaning processes such as radiative transfer, microphysics, and phase changes. In the middle atmosphere there is a three-way interaction between dynamical processes, physical processes (primarily radiative transfer), and chemical processes.

ozone depletion, prompted by the discovery of the Antarctic ozone hole, new satellite observations (e.g., from the Upper Atmosphere Research Satellite), the inclusion of stratospheric levels in numerical weather prediction models and routine meteorological data sets, and the extension of numerical general circulation models to include realistic representation of the middle atmosphere.

The emphasis on ozone and related chemical species has prompted much interest in the transport of chemical species and its representation in models (e.g., Plumb 2002). This review does not attempt to cover the topic of chemical transport (though clearly the transport characteristics of the stratosphere are intimately connected with its dynamics). Furthermore, there is no space in this review to give details of the combination of meteorological and chemical observations that now underpins our current knowledge of the dynamical state of the stratosphere.

The stratosphere, because of the increase in temperature with height, is strongly stably stratified. A convenient density variable, conserved in adiabatic motion, is the potential temperature θ , which increases monotonically upward when the stratification is stable. The dynamics on the large scale is described by theories of rotating stratified flows as developed over the last 60 or so years. Although the large-scale dynamics is now straightforwardly simulated by numerical models, the underlying theory is important in providing a framework in which behavior of models can be understood and their results can be interpreted. There is also significant small-scale, high-frequency motion in the stratosphere and mesosphere. Much of this is associated with inertio-gravity waves, which are forced in the troposphere by flow over topography, convection, shear instability, and sometimes by spontaneous emission from the large-scale flow. These inertio-gravity waves propagate up into the stratosphere and mesosphere where they break and dissipate. This implies a force on the large-scale flow where the breaking or dissipation takes place. There may be additional nondissipative forces due to horizontal refraction (Bühler & McIntyre 2003). Breaking also leads to localized patches of threedimensional turbulence and hence to mixing of chemical species. The length scales of the inertio-gravity waves themselves, let alone the three-dimensional turbulence that arises from their breaking, are well below the spatial resolution of presentday global models, and the effects of inertio-gravity waves on the large scale therefore need to be represented in the numerical models by parametrizations. The whole subject of inertio-gravity waves and their effects is centrally important to understanding and modelling the middle atmosphere and was comprehensively reviewed recently by Fritts & Alexander (2003). It is therefore not covered in any detail here.

The circulation of the middle atmosphere varies strongly in height, latitude, and longitude. However, the most systematic variations are in latitude and height. This motivates a description where at the leading order the longitudinal variation is neglected (by taking the longitudinal average of the observed, threedimensional state, for example). Figure 1 shows height-latitude cross-sections of longitudinal average of longitudinal wind (Randel 1992). There is an important difference between the winter hemisphere, where the flow is dominated



Figure 1 Longitudinally averaged longitudinal component of wind in troposphere and stratosphere for January (Northern Hemisphere winter) and July (Northern Hemisphere summer) (from Randel 1992). Negative regions (westward winds) are shaded. The winter hemisphere has strong eastward jets in the stratosphere (the "polar vortex"). The Southern Hemisphere jet (July) is stronger than the Northern Hemisphere jet (January) because of the weaker wave force in the Southern Hemisphere. (Copyright 1992, W.J. Randel. Reproduced with permission.)

by an eastward "polar vortex," and the summer hemisphere, where the flow is westward.

It has long been known that to explain the longitudinally averaged state it is necessary to take account of the systematic effects of the deviations, usually termed waves or eddies, of the actual circulation from the average or mean. The development of the theory of wave mean-flow interaction in the 1960s and 1970s was significantly stimulated by some of the questions posed by the observed state of the middle atmosphere. This theory is discussed in detail by Andrews et al. (1987) and references therein. One starting point for this review is that the wave mean-flow description continues to provide a useful quantitative framework for understanding the circulation of the middle atmosphere, although some of the limitations of this description are noted.

The molecular diffusivity in the stratosphere increases from 10^{-4} m² s⁻¹ in the lower stratosphere to 10^{-2} m² s⁻¹ in the upper stratosphere and is too small to play any direct role in the dynamics. The patches of three-dimensional turbulence noted above might in principle play a dynamical role, e.g., as a kind of vertical eddy diffusivity. But recent estimates of the mixing effect of these patches from chemical tracer observations (e.g., Legras et al. 2003) suggest a vertical eddy-diffusivity effect of at most 10⁻¹ m² s⁻¹ and probably at least an order of magnitude less, implying that their dynamical role is almost always negligible. A much more important "non-conservative" physical effect is radiative transfer, both short-wave heating primarily associated with ozone, and long-wave transfer, primarily associated in the stratosphere with carbon dioxide, ozone, and, in the very lowest part of the stratosphere, water vapor. The radiative transfer plays both a forcing role, in setting up large-scale pole-equator-pole temperature gradients and associated longitudinal velocity distributions (implied by geostrophic balance), and a dissipative role in that it acts to reduce the amplitude of fluctuations (e.g., wave amplitudes). An estimated thermal damping or relaxation timescale may be used as a measure of this dissipative role is estimated by a thermal damping timescale. This is determined, among other things, by the ambient temperature (smaller temperatures imply increased damping timescales) and by the vertical length scale (smaller scales imply decreased damping timescales). Estimates for vertical length scales of several kilometers based on radiative code calculations are 20-40 days in the lower stratosphere decreasing to 5 days in the upper stratosphere, although it is almost impossible to make such estimates precise (see Zhu 1993, 1997, including details of scale dependence; also Newman & Rosenfield 1997). As noted below, some representation of radiative transfer is an essential ingredient for any model of stratospheric dynamics on longer timescales than a few days.

The emphasis of this review is on the dynamics in the extratropics and its global implications. The topic of mean-flow interaction in the tropics is not treated in detail because one of the most important aspects, the quasibiennial oscillation, was recently reviewed by Baldwin et al. (2001). The structure of the remainder of the review is as follows. Section 2 discusses Rossby waves and stratospheric vortex dynamics. Section 3 considers the dynamics of longitudinally symmetric circulations. Section 4 discusses the extratropical stratospheric circulation, emphasizing the two-way interaction between waves and mean flows. Section 5 discusses dynamical coupling between troposphere and stratosphere (i.e., c, above).

2. ROSSBY WAVES AND STRATOSPHERIC VORTEX DYNAMICS

On the large scale and on timescales greater than a day or so the extratropical stratosphere is well described as a "balanced" system in which there is a single time-evolving scalar field, the potential vorticity (PV), which is materially

conserved in adiabatic, frictionless motion, and from which all other dynamical fields may be instantaneously determined through a PV "inversion" (McIntyre 2003a,b and references therein). In the classical approach of analyzing small-amplitude disturbances to a self-consistent steady flow it is well known that such balanced systems allow, for given spatial structure, a single wave, usually described as a Rossby wave corresponding to the single time derivative in the PV conservation equation. The dynamics of Rossby waves involves horizontal advection (or more precisely, advection along θ -surfaces) of PV (which has a strong pole-to-pole gradient) with resulting changes in temperature and pressure fields and vertical displacement of fluid parcels. The balance assumption excludes other waves, such as inertio-gravity waves and acoustic waves.

An important part of the dynamics of the stratosphere is that planetary-scale Rossby waves (sometimes known as planetary waves) are excited in the troposphere, e.g., by flow over topography, by latent heat release, or through the nonlinear evolution of troposheric eddies (Scinocca & Haynes 1998), and then propagate up from the troposphere into the stratosphere and mesosphere. One important basic result, which may be derived under the rather strong assumptions of smallamplitude waves and background flow varying slowly only in the vertical, is the Charney & Drazin's classic result that waves propagate upward only through flow that is weakly eastward relative to phase speed (with maximum relative flow speed from propagation decreasing as length scale decreases) and only if the scale of the waves is sufficiently large. Given that the dominant forcing of stratospheric Rossby waves is geographically stationary, this provides a basic explanation of why the winter stratosphere (with eastward flow around the pole; see Figure 1) is much more disturbed than the summer stratosphere (with westward flow around the pole) and why the disturbances in the winter stratosphere tend to have much larger scales than is typical of the troposphere below.

The time evolution of the PV field in stratospheric flows gives a clear guide to the dynamical mechanisms operating. McIntyre & Palmer (1983, 1984), using PV maps calculated from observations, distinguished between reversible displacements and distortions of the polar vortex, which they associated with upward propagating Rossby waves, and the nonlinear stirring of the PV field outside the vortex, which they identified with the breaking of those Rossby waves. They called the region outside the vortex the stratospheric "surf zone." Many subsequent modelling and observational studies of the stratosphere further examined this vortex/surfzone structure. Figure 2 shows a recent observational view of the PV field in the stratosphere (Simmons et al. 2004), from an "analysis" data set generated by the European Centre for Medium Range Weather Forecasts (ECMWF). Such data sets are produced by blending new observations with information from old observations that has been carried forward in time by a weather forecast model (in this case the ECMWF model).

The close mathematical relation between three-dimensional balanced systems and the two-dimensional vorticity equation led to the latter being studied as a simple and computationally inexpensive proxy for the three-dimensional system. Starting



PV analysis 12UTC 20 September 2002

Figure 2 (From Simmons et al. 2004). Analyses of PV in the Southern Hemisphere on the 850-K potential temperature surface (corresponding roughly to 30 km), for 12 UTC September 20 and 25 September, 2002. Shading is -1000 PVU (*black*) to 0 (*white*) for potential vorticity (1 PVU = 10^{-6} m² s⁻¹ Kkg⁻¹). The PV distribution on September 20, 2002 shows the polar vortex (large negative values of PV, *dark shading*), surrounded by a "surf zone" in which streamers of air drawn out of the vortex and streamers of air from low latitudes are stirred together. The PV distribution of September 25, 2002 shows the vortex split into two in a sudden warming. This event is unique, in the Southern Hemisphere, over the last 50 years. (Copyright 2004, Am. Meteor. Soc. Reproduced with permission.)

-600

PVU

-1000

-800

-200

-400

with the work of Juckes & McIntyre (1987), a number of numerical studies of twodimensional stratosphere-like flows gave important insights into the dynamics of the stratospheric polar vortex and surf zone, which have provided a useful framework on which to base interpretation of observations and of more realistic modelling studies as three-dimensional simulations have become more feasible. Juckes & McIntyre (1987) noted several important features including material coherence of the vortex (strictly speaking, the high PV core of the vortex), which, even for quite large-amplitude forcing, experienced reversible deformation but with almost no transport of fluid between interior and exterior, and the strong stirring effect of the disturbed flow outside the vortex, which tended to pull filaments of material out of the edge of the vortex and mix them into the exterior flow. They identified the region outside the vortex with the real stratospheric surf zone and noted that the confinement of filamentation and mixing to the exterior of the vortex naturally led to sharpening of the vorticity gradients surrounding the vortex through an erosion process. Another interesting feature of the Juckes & McIntyre simulations was the roll up into coherent small-scale vortices of filaments of highvorticity air that were drawn out of the main vortex. The roll up may, in part, be explained by shear instability, since a PV profile taken across a filament exhibits a local maximum.

Later work with two-dimensional models confirmed this picture and explored the relevant processes in more detail. Some of this work was based on shallowwater models in which a relaxation of the thickness field to a suitable equilibrium distribution was used as a natural analogue of thermal relaxation due to radiative transfer. Juckes (1989) showed how the combined effects of such relaxation and wave forcing could lead to a persistent, sharp-edged vortex (see also Salby et al. 1990) and analyzed the transport characteristics of the flow. Polvani et al. (1995) focused on the subtropical part of the surf zone and showed how its structure depended on the presence of the relaxation term. Chen (1996) further explored the dependence of the structure of the surf zone, particularly its subtropical part, on wind profiles in the tropics and subtropics. The sharp PV gradients that form at the vortex edge through the erosion process motivated a piecewise-constant-vorticity model, with one value of vorticity inside the vortex and another value outside, which could be solved using the contour-dynamics method. This is attractive for reasons of both simplicity and computational economy. Polvani & Plumb (1992) used such a model to examine the process of filamentation and breaking and Polvani & Dritschel (1993) made a more general contour-dynamics-based study of vortex dynamics on a sphere.

Study of these stratosphere-like flows within two-dimensional models naturally focused attention on certain generic problems of two-dimensional vortex dynamics. One is the idealized erosion or "vortex stripping" problem (Legras et al. 2001 and references therein), where the action of a large-scale external flow applied to a preexisting vortex is to remove the outer layers and leave a sharp jump in vorticity between the vortex itself and the exterior. If the external flow is not too strong then the jump eventually becomes large enough to give a dynamical stabilizing effect and hence inhibit the removal of further layers. This is a useful analogue of the action of Rossby wave forcing on the polar vortex, but is relevant to a much more general class of problems, e.g., to the evolution of vortices in two-dimensional turbulence.

A second generic problem is that of the effect of external flow on the evolution of filaments of vorticity. Studies of the stability of one-dimensional filaments show how the action of a strain flow on a filament is stabilizing (Dritschel et al. 1991). The effect of external shear may also be stabilizing (Dritschel 1989). In axisymmetric geometry the effect of the latter is such that a circular filament of vorticity surrounding an isolated vortex of the same sign may be stabilized by the shear of the irrotational external flow. The stabilizing effect is stronger on the sphere (Polvani & Dritschel 1993). These mechanisms imply that the roll up of filaments drawn out of the polar vortex into the surf zone may be relatively rare and are also relevant to the evolution of the filamentary vorticity structure that typically arises outside vortices in two-dimensional turbulent flows.

A basic paradigm for a three-dimensional balanced system is the well-known quasigeostrophic PV equation. This is closely analogous to the two-dimensional vorticity equation with the difference that the flow is layer-wise two-dimensional (in the horizontal plane) and the inversion operator that gives the stream function in terms of the quasigeostrophic PV is an inverse three-dimensional Laplacian-type operator, which is approximately isotropic if vertical distances are divided by Prandtl's ratio of scales f/N, where N is the buoyancy frequency and $f = 2\Omega \sin \phi$, where Ω is the rotation rate of the Earth and ϕ is the latitude, is the Coriolis parameter. The quasigeostrophic system has insufficient accuracy for stratospheric modelling and an effective way of investigating the dynamics of the extratropical stratosphere has been to use so-called primitive-equation models, which in principle allow inertio-gravity waves, but which in practice evolve almost as if balanced when used to simulate large-scale stratosphere-like flows. Concepts such as PV inversion may then be used diagnostically, rather than to predict the evolution of the flow.

Simulations in idealized models, some using the primitive equations (Haynes 1990, O'Neill & Pope 1988, Polvani & Saravanan 2000), some based on threedimensional quasigeostrophic versions of contour dynamics (Dritschel & Saravanan 1994, Waugh & Dritschel 1999), show that in three-dimensional flows, where wave forcing is applied at low levels, there is upward propagation of Rossby waves and the nonlinear evolution of the flow at levels remote from the forcing generally exhibits the pattern identified by Juckes & McIntyre (1987), i.e., with persistence of the vortex as a material entity, strong stirring outside the vortex, which draws filaments of high PV air into the surf-zone, and sharpening of the PV gradients at the vortex edge through the process of vortex erosion. (The interpretation of the nonlinear evolution as Rossby wave breaking is natural in these flows, where there is a clear propagation phase, whereas in the two-dimensional flows described earlier the separation between propagation and breaking is not so clear and hence use of the term "wave breaking" at all is not so clearly justifiable.) Important features that occur in the three-dimensional case include the possibility of breaking at low levels, which destroys the vortex at those levels and prevents propagation of wave activity to upper levels (Dritschel & Saravanan 1994), sensitivity of the behavior to variation in the size and strength of the vortex in the vertical because of the implications for Rossby wave propagation (Polvani & Saravanan 2000, Waugh & Dritschel 1999), and three-dimensionality of the small-scale vortices that form as a result of the roll up of filaments (or, more correctly in the three-dimensional case, sloping sheets) of high PV that are drawn out of the main vortex during wave breaking. Polvani & Saravanan (2000) note that sheets drawn out of the vortex tend to be deep and speculate on the relevance of the vortex alignment process previously identified in quasigeostrophic turbulence.

Although the existence of small-scale vortices in the stratosphere has been argued on fluid-dynamical grounds, this has been very difficult to verify directly from observations. The main source of dynamical information has been from satellite radiometers, which generally have rather coarse resolution in either vertical or horizontal. The ECMWF analysis data set mentioned above is now at very high spatial resolution (40 km in horizontal, 1.5 km in vertical). Figure 3, taken from Simmons et al. (2004), shows pictures of small-scale vortices revealed by this data set. These features in the data set are strongly model-influenced, but Simmons et al. note that they fit well with actual observations. For example, the temperature and wind signatures of one of the small vortices shown on October 4, 2002 is in good agreement with measurements from Australian and neighboring radiosondes.



Figure 3 (From Simmons et al. 2004). Analyses of water vapor concentration (by mass) on the 850-K potential temperature surface (corresponding roughly to 30 km), for 12 UTC October 2 and October 4, 2002. Shading is from 2 mg/kg (*white*) to 3.8 mg/kg (*black*). The water vapor may be regarded as a tracer and high values correspond to high negative values of PV (recall Figure 2) and hence mark air that originated in the vortex. Over the two days shown, the band of high water vapor concentration/high PV air over the Pacific evolves into two separate vortices. (Copyright 2004, Am. Meteor. Soc. Reproduced with permission.)

The lifetime of small-scale vortices such as those seen in Figure 3 is likely limited, by decay due to radiative transfer, to a few days, with deeper vortices lasting longer than shallow vortices. Haynes & Ward (1993) discuss relevant mechanisms and estimates of timescales. Such vortices are also associated with anomalies in chemical species such as water vapor and ozone, and once the vortex itself dissipates, the chemical anomalies are vulnerable to deformation by large-scale shear, stretched into filaments/sheets and ultimately mixed into the background.

If the wave forcing is strong enough, the main vortex may be significantly displaced from the pole, strongly deformed in shape, or even split into two. These events are known to meteorologists as "sudden stratospheric warmings" because they are manifested as very rapid increases in temperature, due to adiabatic warming through descent (as implied by inversion of the evolving PV field). In the Northern Hemisphere sudden stratospheric warmings occur in mid-winter in about half of winters, on average, and there is also often a sudden-warming-like event at the end of winter. Disturbances to the vortex are generally weaker in the Southern Hemisphere than in the Northern Hemisphere, as is expected from the relative lack of topography and greater proportion of ocean versus land, and strong events tend to be confined to the end-of-winter transition-the "final warming"-in October or later. Therefore, there was considerable surprise and interest when, as seen in Figure 2, the Southern Hemisphere polar vortex split into two in September 2002, with a corresponding split of the "ozone hole"—the region of very low ozone air that currently forms annually over the Antarctic in the Southern Hemisphere winter. This event restimulated the interest of the broader atmospheric science community in the dynamics of the stratosphere, particularly in sudden warmings. An issue of the Journal of Atmospheric Sciences dedicated to the September 2002 warming will appear shortly.

The important ingredients of the dynamics of the extratropical winter stratosphere described in this section are upward propagation of Rossby waves on the PV gradients near the edge of the polar vortex and the breaking of these waves at levels where their amplitudes become large, giving rise to filamentation, deformation, and even splitting of the main vortex. In highly disturbed states the role of upward Rossby wave propagation may be small and a vortex-interaction description may then be more relevant.

In the remainder of this review the emphasis is less on vortex dynamics and more on the wave mean-flow description where the flow is divided into a longitudinal average or mean part and a disturbance or wave part. The essence of the theory of wave mean-flow interaction is that there is long-range momentum transport between the location where the waves are generated and the location where the waves break or dissipate (Andrews et al. 1987). The associated force is here called a "wave force." The theory may also be formulated in terms of transport of PV, which is local, within the breaking or dissipation region (e.g., McIntyre & Norton 1990). (The latter alternative may seem the more natural in view of the vortex phenomena described previously.) The relevant theories are most highly developed when the flow is weakly disturbed from a longitudinally symmetric state. Formal expressions for the wave force and wave momentum transport are valid in general, but what is lost when the flow is highly disturbed is any clear association with wave propagation. There is ongoing discussion and investigation of alternative formulations (e.g., see discussion in McIntyre 1982, also Thuburn & Lagneau 1999).

3. THE DYNAMICS OF THE LONGITUDINALLY AVERAGED CIRCULATION

In this section I discuss the dynamics of the longitudinally averaged circulation. As noted above, the systematic effects of waves or eddies are an essential part of this dynamics and are represented here by a longitudinal wave force \mathcal{G} per unit mass. This exploits the simplification allowed either by the so-called "transformed Eulerian mean formalism" or by formulating the dynamical equations using θ as a vertical coordinate (e.g., Andrews et al. 1987). \mathcal{G} represents the effect of waves, such as Rossby waves and gravity waves, on the mean flow. In the context of Rossby waves, \mathcal{G} has to represent the propagation, breaking, and vortex interaction behavior described in the previous section and therefore has to be a complicated nonlinear, (and, as yet, undetermined) function of the mean flow and of wave sources. Gravity wave propagation and breaking also depends strongly on the background flow (Fritts & Alexander 2003) and has to be incorporated into \mathcal{G} . In this section we consider the much more straightforward problem of how the circulation responds to a given force.

3.1. Basic Principles

A suitable simplified set of model equations for this problem is as follows:

2

$$\frac{\partial \bar{u}}{\partial t} - 2\Omega \sin \phi \bar{v}^* = \mathcal{G} - \gamma \bar{u}$$
(3.1)

$$\Omega \sin \phi \frac{\partial \bar{u}}{\partial z} + \frac{R}{aH} \frac{\partial \bar{T}}{\partial \phi} = 0$$
(3.2)

$$\frac{\partial \bar{T}}{\partial t} + \bar{w}^* \left(\frac{HN^2}{R}\right) = \bar{Q}_s + \bar{Q}_l(\bar{T}) = \bar{Q}_s - \alpha \bar{T}$$
(3.3)

$$\frac{1}{a\cos\phi}\frac{\partial}{\partial\phi}(\bar{v}^*\cos\phi) + \frac{1}{\rho_0}\frac{\partial}{\partial z}(\rho_0\bar{w}^*) = 0$$
(3.4)

Here ϕ is latitude and z is a log-pressure coordinate. *a* is the radius of the Earth, *R* is the gas constant for dry air, *H* is the constant nominal density scale height used to define the log-pressure coordinates and ρ_0 is the background density in these coordinates, equal to $e^{-z/H}$. *H* is usually taken to be about 7 km, implying rough correspondence between the log-pressure coordinate *z* and geometric altitude. The independent variables are \bar{u} (longitudinal velocity), \bar{T} (temperature), and (\bar{v}^* , \bar{w}^*) (latitudinal and vertical components of the circulation in the latitude-height plane,

usually called the mean meridional circulation). Equations 3.1–3.4 are strictly valid only in the limit of small disturbances to a resting basic state in which the temperature varies only in height and for slow time variations (i.e., longer than a day or so). Equations 3.1–3.4 define a balanced system in the sense discussed in Section 2. (There are apparently two time derivatives, but one may be eliminated.)

The notation (.) emphasizes that the variables may be interpreted as longitudinal averages, with the meridional circulation taken as the "transformed Eulerian mean" circulation, in which case the effect of the eddies enters only through the wave force \mathcal{G} in the longitudinal momentum equation (Equation 3.1). Following tradition, the term $\gamma \bar{u}$ is included in this equation to represent linear friction, but it is very difficult to argue that such a friction is at all relevant in the stratosphere. The second term on the left-hand side represents the Coriolis torque associated with the mean meridional circulation. Equation 3.2 expresses the so-called thermal wind relation between longitudinal velocity and temperature which follows from geostrophic and hydrostatic balance.

In the temperature equation (Equation 3.3), N^2 is the square of the buoyancy frequency, defined as

$$N^2 = \frac{R}{H} \left(\frac{dT_0}{dz} + \frac{\kappa T_0}{H} \right),$$

where T_0 is a reference temperature dependent only on z and $\kappa = R/c_p \simeq \frac{2}{7}$, where c_p is the specific heat of dry air at constant pressure. \bar{Q}_s and \bar{Q}_l are the short-wave and long-wave radiative heating rates, respectively. The latter is approximated as $-\alpha \bar{T}$, which is an extreme simplification but, as discussed above, captures the fact that the real radiative transfer is essentially relaxational. The final equation (Equation 3.4) expresses mass continuity for the meridional circulation.

Equations 3.1–3.4 describe the coupled response of the wind and temperature fields on the one hand and the meridional circulation in the latitude-height plane on the other, to applied wave force \mathcal{G} and short-wave heating \bar{Q}_s . Any three independent variables can be eliminated to give a single equation for the fourth. The discussion below focuses on the meridional circulation, in particular the vertical velocity \bar{w}^* , but once this is known other fields can be deduced as required. To highlight the dependence of the response on the spatial structure and the time dependence of \mathcal{G} and \bar{Q}_s , it is instructive to consider the case where they are assumed to be sinusoidal functions of time with given frequency σ , i.e., $\mathcal{G} = \text{Re}(\hat{\mathcal{G}}e^{i\sigma t})$ and $\bar{Q}_s = \text{Re}(\hat{Q}_s e^{i\sigma t})$, following Garcia (1987). It follows, by combining Equations 3.1–3.4, that the vertical velocity $\bar{w}^* = \text{Re}(\hat{w}e^{i\sigma t})$ is the solution of the equation

$$\frac{\partial}{\partial z} \left[\frac{1}{\rho_0} \frac{\partial(\rho_0 \hat{w})}{\partial z} \right] + \left(\frac{i\sigma + \gamma}{i\sigma + \alpha} \right) \frac{N^2}{4\Omega^2 a^2 \cos\phi} \frac{\partial}{\partial\phi} \left[\frac{\cos\phi}{\sin^2\phi} \frac{\partial \hat{w}}{\partial\phi} \right]$$
$$= \frac{1}{2\Omega a \cos\phi} \frac{\partial}{\partial\phi} \left[\frac{\cos\phi}{\sin\phi} \frac{\partial\hat{\mathcal{G}}}{\partial z} \right] + \left(\frac{i\sigma + \gamma}{i\sigma + \alpha} \right) \frac{R}{4H\Omega^2 a^2 \cos\phi} \frac{\partial}{\partial\phi} \left[\frac{\cos\phi}{\sin^2\phi} \frac{\partial\hat{\mathcal{Q}}_s}{\partial\phi} \right]$$
(3.5)

Note first that the operator on the left-hand side is elliptic, so that the vertical velocity response to a forcing localized to a region spreads away from that region. This is as expected from PV inversion—the flow is assumed balanced and the non-local response to the applied forcing may be deduced by applying the appropriate inversion operator to the PV field, which is changing in time because of the applied forcing. Also note the fact that the different inverse timescales in the problem, the frequency σ and the frictional and thermal relaxation rates γ and α , respectively, appear only in the factor $(i\sigma + \gamma)/(i\sigma + \alpha)$ that appears both on the left-hand side and right-hand sides of Equation 3.5. Having displayed this dependence, the value of γ is set to zero, for reasons given above.

In the high-frequency limit, when thermal relaxation is negligible, the factor $i\sigma/(i\sigma + \alpha)$, which appears in two places in Equation 3.5, is approximately 1. The form of the operator on the left-hand side then implies that the response tends to have aspect ratio of the order of Prandtl's ratio of scales, f/N or $2\Omega \sin \phi/N$, mentioned above, with the structure of the response shallowing near the equator. The short-wave heating term on the right-hand side can be significant, implying that both wave force and short-wave heating may be effective in driving a mean circulation. As frequency decreases toward zero relative to the thermal damping rate, the modulus of the factor $i\sigma/(i\sigma + \alpha)$ also decreases towards zero. This has two important effects. First, the heating term on the right-hand side diminishes in importance. Second, the aspect ratio of the response changes, so that when $\sigma \ll \alpha$ the characteristic ratio of vertical to horizontal scales is $(2\Omega \sin \phi/N)(\alpha/\sigma)^{1/2}$, i.e., the response to given \mathcal{G} deepens in the vertical (or narrows in latitude).

In the steady-state limit $\sigma/\alpha \rightarrow 0$ the short-wave heating term vanishes, implying that such heating drives no meridional circulation and is precisely balanced by the thermal relaxation, so that the right-hand side of Equation 3.3 vanishes. Furthermore, that second term on the left-hand side of Equation 3.5 also vanishes, so that the vertical velocity at a given latitude is driven only by the applied force (or, more strictly, the latitudinal derivative of the force) at that latitude. Vertical integration of Equation 3.4 in this limit, substitution from Equation 3.1, and application of the requirement that the vertical velocity *w* stays finite as $z \rightarrow \infty$, gives the result (Haynes et al. 1991)

$$\bar{w}^* = -\frac{1}{2\Omega\rho_0(z)\cos\phi}\frac{\partial}{\partial\phi}\left[\frac{\cos\phi}{\sin\phi}\int_{z}^{\infty}\rho_0(z')\mathcal{G}(\phi,z')dz'\right],\tag{3.6}$$

implying that, in the steady-state limit, the vertical velocity at a given level is determined only by the wave force directly above that level. Figure 4, taken from Holton et al. (1995), schematically shows the variation between the high-frequency response and the steady-state response.

The transition between the time-dependent and steady limits depends on the spatial structure and location of the force. Assuming that the height scale is Δz and the latitudinal scale is $\Delta \phi$, the estimate of the two terms on the left-hand side of Equation 3.5 implies that the transition between the two limits takes place when



Figure 4 (From Holton et al. 1995). Idealized numerical experiments on the meridional circulation reponse of the stratosphere to a longitudinally symmetric westward force applied in the shaded region. Contours are streamlines, with the same contour interval used in each panel (and corresponding to a given mass flux). (*a*) Adiabatic response for $\sigma/\alpha \gg 1$. A smaller contour interval in this case would show an oppositesigned circulation cell above the shaded region and extension of the lower circulation cell well into the opposite hemisphere. (*b*) Response for $\sigma/\alpha \simeq 0.34$, corresponding to annual frequency and 20-day radiative damping timescale; the solid and dashed contours show the response that is respectively in phase and 90° out of phase with the forcing. (*c*) Steady-state response ($\sigma/\alpha \ll 1$). Note that the magnitude of the response in the mass stream function increases as σ/α decreases and that it is given qualitatively by Equation 3.6 for (*c*). The model is unbounded below so the ordinate has arbitrary origin. (Copyright 1995, Am. Geophys. Union. Reproduced with permission.)

$$\sigma \sim \frac{4\Omega^2 a^2 \sin^2 \phi(\Delta \phi)^2}{N^2 \min\{\Delta z^2, \Delta z H\}} \alpha.$$
(3.7)

It follows that the timescale required for the steady-state limit to be achieved is longer at low latitudes. Near the equator, the $\sin^2 \phi$ should be replaced by $(\Delta \phi)^2$ and rearranging, it follows that for given frequency σ , the steady-state limit is not achieved in a range of latitudes $\Delta \phi_{\sigma}$ given by

$$\Delta\phi_{\sigma} \sim \left(\frac{N^2 \min\{\Delta z^2, \Delta z H\}}{4\Omega^2 a^2}\right)^{1/4} \left(\frac{\sigma}{\alpha}\right)^{1/4}$$
(3.8)

(Holton et al. 1995).

3.2. Implications for the Mean Meridional Circulation

The sense of the circulations shown in Figure 4 corresponds to a westward force, as is expected to be present in the winter hemisphere of the real stratosphere, as a result of breaking and dissipating Rossby waves. In the high-frequency limit the response to a localized force extends significantly in latitude. This is manifested in the observed response to large wave force, e.g., associated with sudden warming events. It has been known for some time that at times of significant westward wave force, which drives downward motion and hence warming at high latitudes, there is an opposite response at low latitudes and in the opposite hemisphere, with observed cooling that may be explained by the compensating upward motion (e.g., Randel 1993).

The dependence of the response on σ/α shown in Figure 4 has important implications for the seasonal variation of the mean circulation and for the longtime mean circulation. The annual variation is of particular interest. Yulaeva et al. (1994) argued that the well-known annual cycle in tropical lower stratospheric temperatures, with cold temperatures in Northern Hemisphere winter and warm temperatures in Northern Hemisphere summer, was likely due to the annual cycle in the extratropical wave force, with relatively large force in Northern Hemisphere winter compared to Southern Hemisphere winter (for reasons discussed in section 2). For the annual frequency the ratio $\sigma/\alpha \simeq 0.3$ (corresponding roughly to panel b of Figure 4), so it is important to consider the reduction in latitudinal penetration of the meridional circulation as σ/α decreases. On this basis Holton et al. (1995) argue that the wave force relevant for the annual cycle is likely subtropical rather than extratropical. (Equation 3.8 gives an estimate of the relevant range of latitudes.) Randel et al. (2002) made a detailed study of the time variation of tropical upwelling and the relation to extratropical wave force and suggest that the annual cycle response is particularly large because the radiative relaxation timescale α^{-1} in the tropical lower stratosphere is very long (perhaps 100 days). This is consistent with some aspects of their results, but implies a significant difference between the phase of the maximum upward velocity and the phase of the maximum temperature and seems at odds with other observational studies, e.g., Rosenlof (1995). The overall

plausibility of the Yulaeva et al. (1994) hypothesis (with the "subtropical" caveat) was demonstrated by Scott (2002) in simulations with an idealized "mechanistic" model of the stratosphere including seasonal variation. (Such models are discussed in more detail in section 4). With no forcing of Rossby waves in the extratropics, and hence no wave force, there is no annual and only a weak semiannual cycle in tropical lower stratospheric temperatures. With forcing of Rossby waves only in one hemisphere (a simple representation of the real interhemispheric difference), there is a significant annual cycle of temperature with the expected phase. However, Plumb & Eluszkiewicz (1999) point out that the observed vertical velocities, which tend to maximize on the summer side of the equator, are not wholly explained by the response to wave forces, and argue that heating, i.e., the annual variation of \bar{Q}_s in Equation 3.3, also needs to be considered, supporting their arguments with simple numerical simulations.

The mean meridional circulation is particularly important on long timescales because of its role in transporting chemical species in the vertical across θ surfaces. In the tropical lower stratosphere there is systematic year-round upwelling, which plays an important role in carrying species with sources in the troposphere, many crucial for ozone chemistry, up into the stratosphere (e.g., WMO 1999, 2003). Considering long timescales naturally focusses attention on the steady-state limit, $\sigma/\alpha \rightarrow 0$, in which the result Equation 3.6 holds, as illustrated by Figure 4c. This result is discussed in detail by Haynes et al. (1991), who describe it as a "downward control principle" for the effect of wave force \mathcal{G} on the mean meridional circulation. It is important to realize the limitations of this result-the downward control puts no constraint on what caused the wave forces and, in the case of most such forces in the stratosphere, the waves responsible have propagated up from the troposphere—but, nonetheless, it provides useful insights into the workings of the mean meridional circulation. For example, one implication is that forces exerted high in the middle atmosphere, e.g., in the upper stratosphere and mesosphere, may be important in determining the vertical velocity and, hence, temperatures in the lower stratosphere. In practice, the density factor within the integral works against this, but such forces may be important if wave forces in the lower and middle stratosphere are very small, as they may be, for example, in Southern Hemisphere mid-winter. (See further discussion in section 4.) Garcia & Boville (1994) demonstrate an important role for mesospheric gravity wave forces in this case.

The result, as expressed by Equation 3.6, that in the steady-state limit the meridional circulation is driven only by wave forces, not by heating (i.e., \bar{Q}_s), has important implications for understanding the time-averaged mean meridional circulation in the stratosphere. The term "extratropical pump" is sometimes used to describe the action of wave forces on the meridional circulation (Holton et al. 1995), with the emphasis on the extratropics due, in part, to the fact that the wave force is apparently strongest there. But, as Plumb & Eluszkiewicz (1999) note, great care is needed in distinguishing tropics from extratropics for this purpose. As implied by the sin ϕ factor in the denominator of Equation 3.6, small wave forcing close to the equator could be highly significant.

However, a more fundamental point here is that the assumptions leading to Equation 3.6 cannot be justified close the equator. First, the linearization about a state of rest neglects (in Equation 3.1) the latitudinal (and vertical) gradients of the mean longitudinal flow u, relative to the Coriolis parameter. Equation 3.6 may be generalized appropriately (Haynes et al. 1991) and implies control that is downward along angular momentum surfaces rather than exactly downward. This allows the possibility of latitudinal, as well as vertical, penetration of the effects of wave forces, particularly near the equator (Dunkerton 1991, Tung & Kinnersley 2001), although this penetration is not likely very important in the lower stratosphere (Plumb & Eluszkiewicz 1999). Nonetheless, Semeniuk & Shepherd (2001b) demonstrate some latitudinal penetration of the effects of a westward force applied in the subtropics. They also explore the possibility of a middle atmosphere Hadley circulation in the sense of Held & Hou (1980), meaning a region where angular momentum is constant and a steady thermally driven circulation is therefore possible without any wave forcing, as suggested by Dunkerton (1989). Semeniuk & Shepherd (2001a,b) find that in the upper stratosphere, but not in the lower stratosphere, such a circulation leads to significant time-averaged upwelling.

Second, if there is weak frictional relaxation then this must inevitably become important near the equator. For example, with the frictional relaxation term $\gamma \bar{u}$ retained, the Equations 3.1–3.4 imply, in the steady state if $\gamma \ll \alpha$, a thin equatorial layer of thickness

$$\Delta\phi_{\gamma} \sim \left(\frac{N^2 \min\{\Delta z^2, \Delta z H\}}{4\Omega^2 a^2}\right)^{1/4} \left(\frac{\gamma}{\alpha}\right)^{1/4}$$
(3.9)

(analogous to the $\Delta \phi_{\sigma}$ in Equation 3.8) in which frictional relaxation enters the dynamical balance and, in particular, allows a meridional circulation without any local wave force (Plumb & Eluszkiewicz 1999). The 1/4-power dependence on γ makes it essentially impossible to exclude frictional effects from low latitudes in numerical models.

More generally, again as discussed by Plumb & Eluszkiewicz (1999), one key problem of this approach in analyzing the forcing of the mean circulation near the equator is that the wave force \mathcal{G} is treated as given. This is justifiable in the extratropics where the change in velocity due to the applied forcing is relatively small. (It may be calculated via change in temperature implied by the steady state form of Equation 3.3 and thermal wind balance Equation 3.2.) However, the change in velocity increases as the equator is approached and, in the relevant time-dependent problem, a significant part of the wave force must directly drive a change in the mean flow (Haynes 1998). Given that the spatial distribution of the wave force likely depends strongly on any changes to the mean flow, it follows that a nonlinear problem must be solved in which wave force and change in mean flow are deduced together. Similar considerations apply in the subtropical upper troposphere (Held & Phillips 1990). In the mechanistic model simulations reported by Scott (2002) there is a self-consistent calculation of both the waves (and hence \mathcal{G}) and the meridional circulation response. The small amount of wave force within 10° or so of the equator appears to play an important role in the momentum balance that allows a broad tropical upwelling, but arises naturally from the breaking and dissipation of the Rossby waves represented in the model. It may well be the systematic westward sign of a Rossby-wave-induced \mathcal{G} and the inevitability of some small penetration of such \mathcal{G} to very low latitudes that accounts for the apparent robustness of the real tropical upwelling.

In summary, the theoretical approach of taking the wave force as specified suggests that, in principle, it would be possible for the force to stop abruptly at say, 20° with, by Equation 3.6, no upwelling equatorward of that. But this situation is not likely relevant to the real atmosphere. The realistic behavior of breaking and dissipating gravity waves and, above all, Rossby waves likely leads to small wave forces at low latitudes.

4. WAVE MEAN FLOW INTERACTION IN THE EXTRATROPICAL STRATOSPHERE

We now consider the implications of combining the longitudinally symmetric dynamics reviewed in section 3 with Rossby wave propagation, as discussed in section 2. The Equations 3.1–3.4 are still relevant, but with the important extra ingredient over the discussion in section 2 that the wave force G now depends on the \bar{u} and \bar{T} fields because the wave propagation, breaking, and dissipation depend on the background state. Important guiding principles are that Rossby wave propagation is generally inhibited by westward or strong eastward winds \bar{u} and enhanced by strong PV gradients. A relevant model must have some kind of radiatively determined temperature field $T_r(\phi, z, t)$ toward which temperature field is relaxed by long-wave radiative transfer. In Equation 3.3 this would be achieved by choosing $Q_s(\phi, z, t) = \alpha T_r(\phi, z, t)$. The time dependence of T_r allows representation of seasonal variation. The question to address is what dynamical ingredients are needed to account for the observed seasonal and interannual variation.

A natural simplification is to consider models that are "stratosphere-only" (though in practice to avoid problems with artificial upper boundaries such models usually extend into the mesosphere or higher). However, the effect of tropospheric wave sources must be included. Therefore, stratosphere-only models are usually formulated with an artificial lower boundary, e.g., at about 10 km, and with some representation of tropospheric sources by forcing at that boundary, e.g., through an artificial topography. Such models are often described as mechanistic models. Interhemispheric differences might be captured in such models by comparing large wave forcing (Northern Hemisphere) with small or moderate wave forcing (Southern Hemisphere). A next stage of sophistication is to use a general circulation model (GCM) with a dynamically active troposphere.

In studying the stratospheric circulation it is natural to emphasize two extreme possibilities (Yoden et al. 2002). One is where the stratospheric circulation is

governed entirely by the troposphere. In this view dynamical events such as sudden warmings arise as a result of increased wave forcing from the troposphere. In a stratosphere-only model such an event would naturally be initiated by increasing the amplitude of the wave forcing at the artificial lower boundary. The opposite extreme is that of "independent stratospheric variation" in which, in the presence of nonzero but constant tropospheric wave forcing, the nonlinear internal dynamics of the stratosphere give rise to dynamical events such as sudden warmings. Reality appears to be somewhere in between.

Both extremes involve a two-way interplay between waves and mean flow in the extratropical stratosphere, as was first made clear using models with a restricted spatial representation of the different dynamical fields. In particular, Holton & Mass (1976) formulated a quasigeostrophic model in a mid-latitude channel in which (*a*) only a single Fourier mode was used to represent the spatial structure across the channel and (*b*) only the longitudinal mean and a single longitudinal wave mode were retained. The evolution equation reduced to two coupled partial differential equations in *z* and *t*, one describing the effect of the wave force \mathcal{G} on the mean flow and the other describing the effect of the mean flow on the wave propagation and the force \mathcal{G} . A model incorporating the approximations (*a*) and (*b*) is referred to as a Holton-Mass model.

Holton & Mass (1976) demonstrated for steady radiative conditions (i.e., T_r independent of t) that for small wave forcing at the lower boundary a steady state was possible, whereas for larger forcing the interaction between waves and mean flow led to a vacillating or oscillating state, with the oscillation manifested both in the mean-flow strength and the wave propagation. The Holton-Mass vacillations are a simple example of oscillations arising from two-way interaction between waves and mean flow for Rossby waves in the extratropics (with the relaxation to the radiatively determined state playing an important role) and in that sense are the extratropical analogue of Plumb's simple model of the tropical quasibiennial oscillation (Plumb 1977). Holton & Mass argued that sudden warmings might be interpreted as the manifestation of vacillations arising from the internal dynamics of the stratosphere (for fixed but sufficiently large wave forcing). Yoden's (1987) detailed mathematical analysis of the Holton-Mass model revealed the bifurcation structure of the model and clearly revealed the possibility of multiple steady states and bifurcations from a steady state to an oscillating state, and later analysis by Christiansen (2000) revealed further bifurcations leading to chaotic solutions in some parameter regimes.

Many recent studies suggest that the phenomena exhibited by the Holton-Mass model can be identified in more realistic models. Scaife & James (2000) and Scott & Haynes (2000) demonstrated the existence of multiple steady states and vacillations in stratosphere-only models, with fixed lower-boundary wave forcing, and noted the role of latitudinal, as well as vertical, wave propagation. Boville (1986) gave some indication of multiple steady states, one a strong-flow, weak-wave state and the other a weak-flow, strong-wave state, in a perpetual-January GCM simulation including a realistic troposphere. Christiansen (1999) also showed

stratospheric vacillations in a GCM and furthermore demonstrated that they were independent of tropospheric variability and disappeared when the Northern Hemisphere topography was removed, consistent with the Holton & Mass finding that the vacillations were present only for sufficiently strong tropospheric wave forcing.

A natural next stage is to consider stratosphere-only models in which the lowerboundary wave forcing remains fixed, but a seasonal cycle is imposed in the relaxation temperature field. The Holton-Mass model was investigated in this context by Yoden (1990), who argued that a quasisteady analysis provided a useful qualitative guide, but not a complete guide, on which to base interpretation of the time-dependent evolution, and showed that for moderate-amplitude wave forcing the flow began the winter in a weak-flow, strong-wave regime and then made a rapid transition to a strong-flow, weak-wave regime. Near the end of winter there was corresponding reverse transition. This captured some aspects of the observed Southern Hemisphere behavior. For large-amplitude forcing the flow throughout the winter remained in the weak-flow, strong-wave regime, with associated vacillations, a simple representation of Northern Hemisphere behavior. (Recall the interhemispheric differences shown in Figure 1). Scott & Haynes (2002), following on earlier studies by Holton and Wehrbein (1980, 1981), conducted a similar study in a model with latitudinal structure. This model also exhibited strong differences between moderate wave forcing, where the mean flow remained close to the radiative equilibrium flow throughout the winter, and strong wave forcing, where sudden warmings take place in mid or even early winter (earlier with larger wave forcing). Scott & Haynes also showed that two different late winter states, one with strong flow, one with weak flow, were possible for the same value of late winter wave forcing, and were achieved by different early winter forcing.

A different angle on the idea of multiple states and sensitivity of evolution is provided by Gray et al. (2003), who carried out mechanistic model integrations with fixed lower boundary wave forcing, applying radiative conditions for perpetual mid-winter. For each value of wave forcing the model was integrated from many different initial conditions, all corresponding to late summer, but differing in small details. For small forcing amplitudes all cases tended toward a strong flow state. For large forcing amplitudes all cases gave repeated warmings and hence exhibited a weak flow state. A range of intermediate forcing amplitudes showed considerable variation from one initial condition to the other, some giving sudden warmings and others (over a period of 200 days or so) remaining in a strong flow state.

The mechanistic models also suggest the possibility of interannual variability of the stratospheric circulation arising through the internal dynamics of the stratosphere (i.e., with lower-boundary wave forcing and radiation that repeat identically each year). Even in the Holton-Mass model, the chaotic behavior found by Christiansen (2000) is almost certain to imply interannual variability if an annual seasonal cycle is included. A clear mechanism for interannual variability in a model with unconstrained latitudinal structure was found by Scott & Haynes (1998), who demonstrated, for an intermediate range of values of wave forcing, a spontaneous biennial oscillation in which weak wave, strong flow and strong wave, weak flow winters alternated. An important part of the mechanism was the long damping timescale for \bar{u} anomalies at low latitudes, arising from the weakness of the associated \bar{T} anomalies (consistent with the reasoning in section 3 leading to Equation 3.8) endowing the low-latitude \bar{u} field with an interannual memory, like a "flywheel."

In the mechanistic models just described the wave forcing is applied as a specified perturbation at the artificial lower boundary, but the flux of wave activity through the lower boundary varies strongly (e.g., Dunkerton et al. 1981, Gray et al. 2003, Scott & Haynes 2002), almost certainly as a result of partial backreflection of Rossby waves within the stratosphere. Some aspects of the behavior may be a consequence of the artificial lower boundary condition, but the presence of vacillation behavior in models that include a troposphere (Christiansen 1999, Scott & Polvani 2004) suggests that, in reality, it is possible for the stratosphere to play a role in determining the net wave flux it receives from the troposphere.

The interplay between dynamical variability imposed by the troposphere and the two-way interaction between waves and mean flow in the stratosphere was explored by Taguchi et al. (2001) and Taguchi & Yoden (2002a,b) in a series of simulations with a simplified GCM. Large-scale Rossby waves were generated both directly by a simple topography and also through nonlinearity by baroclinic eddies in the model troposphere. In perpetual January conditions large-amplitude topographic wave forcing gave a vacillation regime in the stratosphere, though, because of the highly variable tropospheric circulation, the vacillations were more irregular than typically seen in mechanistic models (Taguchi et al. 2001). With an imposed seasonal cycle (Taguchi & Yoden 2002a) weak/intermediate topographic wave forcing gave significant interannual variability only in late winter (typical of Southern Hemisphere), whereas stronger forcing gave strong interannual variability throughout the winter (typical of Northern Hemisphere). Taguchi & Yoden (2002b) examined the year-to-year interannual variability more carefully and showed that there was no indication of any year-to-year memory, suggesting that the mechanism for interannual variability described by Scott & Haynes (1998) is overwhelmed by the effect of tropospheric variability.

The overall picture from mechanistic-model and GCM studies is that the interplay between Rossby waves and mean flow in the winter stratosphere implies both sensitivity and variability. The sensitivity is not of the precise form implied by the simplest models of the Holton-Mass type, but is important nonetheless and strongly modulates the variability imposed by the troposphere. The sensitivity of the dynamical behavior plays a role in the "cold pole" problem of numerical models of the stratosphere, where errors in radiation schemes or the absence of gravity wave drag can lead to very large errors in the circulation (e.g., Boville 1995). The variability, particularly in view of evidence for natural variability of the coupled troposphere-stratosphere system on timescales out to at least a decade (e.g., Butchart et al. 2000), makes it very difficult to identify systematic change (e.g., due to radiative effects of changes in stratospheric ozone purely from observations) and implies that any numerical modelling study requires long integrations or large ensembles to ensure statistical significance (WMO 2003).

5. STRATOSPHERE-TROPOSPHERE COUPLING

Traditionally, the dynamical coupling between the troposphere and the stratosphere has been considered a one-way effect of the troposphere on the stratosphere, with the troposphere acting as a source of Rossby waves and gravity waves, which propagate upward into the stratosphere. However, observational and modelling evidence is now so strong that the stratosphere does not always play a passive role in the coupled dynamics. Indeed, it may sometimes be appropriate to regard changes in the stratosphere as causing changes in the troposphere.

Much of the evidence focuses on the Northern Hemisphere and Southern Hemisphere annular modes (NAMs and SAMs, respectively) (e.g., Thompson & Wallace 2000), which are dominant signals in variability in the troposphere and believed to arise primarily from two-way interaction between baroclinic eddies and the tropospheric mid-latitude jet (e.g., Feldstein & Lee 1998, Lorenz & Hartmann 2003, Robinson 1991). In Northern Hemisphere winter there is significant correlation between the annular mode index defined on the basis of the surface pressure field and the stratospheric circulation, so that, for example, when there is a strong pole-to-equator pressure gradient at the surface, indicating strong eastward surface flow, there are also strong eastward winds throughout the mid-latitude troposphere and in the mid-to-high latitude stratosphere (Thompson & Wallace 2000). There is corresponding organization in the wave fluxes, indicating propagation and momentum transport, both in the troposphere (Limpasuvan & Hartmann 2000) (as is expected from the accepted mechanism for the variability) and also in the stratosphere (Hartmann et al. 2000) (see Figure 5), implying an associated correlation in the wave force \mathcal{G} in the stratosphere.

Baldwin & Dunkerton (1999, 2001) showed that the vertical structure of NAM variation in Northern Hemisphere winter typically shows a downward progression from middle stratosphere to troposphere (see Figure 6). Does Figure 6 imply a direct effect, with some delay, of anomalies in the circulation in mid-stratosphere on the troposphere?

There are several possible mechanisms by which the stratosphere might affect the troposphere. One possible mechanism is via the nonlocal PV inversion operator. Any change in the PV in the lower stratosphere will instantaneously give rise to changes in wind and temperature in the troposphere. Hartley et al. (1998), Ambaum & Hoskins (2002), and Black (2002) show explicit calculations to illustrate this point. This might account for the lower part of the pattern in Figure 6. A precisely equivalent statement, applied to longitudinal mean fields, is that a wave force \mathcal{G} localized to the lower stratosphere will, through the instantaneous induced meridional circulation, give rise to an acceleration in the troposphere below.



Figure 5 (From Hartmann et al. 2000.) Composites for periods of high and low NAM index and their difference (*left, center*, and *right*, respectively) in longitudinal wind (*top*) and Eliassen Palm flux, which indicates wave propagation and transport of westward momentum, and its divergence, which indicates eastward wave force. Positive contours are gray, negative contours are black, and negative regions are shaded. The Eliassen Palm flux is calculated only for longitudinal wavenumbers 1, 2, and 3. In the "high" phase wave fluxes tend to be directed equatorward within the troposphere and to converge in the subtropical troposphere, whereas in the "low" phase wave fluxes tend to be directed upward from troposphere to stratosphere and to converge, implying an anomalous westward wave force, in the mid- and high-latitude stratosphere. (Copyright 2000, Natl. Acad. Sci, U.S.A. Reproduced with permission.)

Furthermore, as noted in section 3, on longer timescales in the presence of radiative damping, the meridional circulation tends to be narrower and deeper below, potentially allowing an enhanced tropospheric response to a stratospheric wave force. Song & Robinson (2004) suggest that the downward penetrating response in the mean circulation communicates the effect of stratospheric wave forcing to the troposphere, where the response is amplified by the eddy (i.e., wave) feedbacks associated with annular variability.

A different mechanism for communication in the vertical is via Rossby wave propagation. Thus, one interpretation, probably the most obvious interpretation, of the patterns shown in Figure 5 is that upward propagating Rossby waves communicate the effect of annular mode changes in the troposphere to the stratosphere. But the changes in the wave flux might be caused by variation in the refractive properties of the lower stratospheric flow (Hartmann et al. 2000, see also Limpavusan & Hartman 2000), or by downward reflection from higher in the stratosphere (e.g., Perlwitz & Harnik 2003). Song & Robinson (2004) found that the effect of stratosphere on troposphere in their model simulations is significantly reduced when planetary waves are damped in the stratosphere and on this evidence rejected their original hypothesis of downward penetration of the mean circulation and argued



Figure 6 (From Baldwin & Dunkerton 1999.) Correlations between the timeseries at each level of a measure of the NAM with that at 10 hPa. (10 hPa corresponds roughly to 30 km, 100 hPa to 15 km.) There is clear downward progression of the correlation in the stratosphere, but this may be "phase propagation" rather than propagation of information. (See text for further details.) (Copyright 1999, Am. Geophys. Union. Reproduced with permission.)

that Rossby waves likely play a significant role in downward communication of information.

A further possible mechanism for downward communication of information is two-way interaction between waves and mean flow. However, Plumb & Semeniuk (2003) demonstrated that downward patterns similar to the upper part of the Baldwin & Dunkerton pattern shown in Figure 6 (which are also similar to those seen in the vacillation behavior described in section 4) arise naturally, in a Holton-Mass type model, as a response to forcing in the lower stratosphere. Indeed, in simple one-dimensional models of the equatorial quasi-biennial oscillation, which exhibit similar downward progressing features, one may argue with certainty that no downward propagation of information is possible (Plumb 1977). This argument is possible because of the neglect of rotation (the Coriolis parameter is zero at the equator), implying a local relation in the vertical between wave forcing and acceleration and the assumption that the waves may be modelled using the slowly varying approximation. In the extratropical stratosphere neither of these assumptions can be justified—in particular the relevant waves are Rossby waves, which have large vertical wavelengths and for which the slowly varying approximation is not justified. The Plumb & Semeniuk (2003) evidence notwithstanding, this leaves the possibility that there is some real downward causal influence. Indeed, numerical experiments that investigate the effect of perturbations in the upper stratosphere, representing solar variability effects, on the extratropical stratospheric circulation (e.g., Gray 2003, Kodera et al. 1990) in many cases show that the effects of those changes propagate downward and are significant in the lower stratosphere.

Other evidence that the stratosphere plays an active, rather than a passive, role in tropospheric variations associated with the NAM comes from observed and modelled changes in the timescale of these variations. Baldwin et al. (2003) showed that this timescale is significantly longer at times of the year (Northern Hemisphere winter, Southern Hemisphere spring) when there is strong Rossby wave propagation into the stratosphere. Correspondingly, artificial suppression of stratospheric variability in model simulations reduces the timescale of the tropospheric NAM (Norton 2003). The dynamical mechanism here is likely that, when there is significant flux of Rossby waves into the stratosphere, the flow in the stratosphere acts as an integrator (and hence low-pass-filter) of this flux (or rather the variability in this flux) because stratospheric damping times are relatively long. Any stratospheric effect on the troposphere therefore tends to increase the timescales of the variability. When there is little flux of Rossby waves into the stratosphere (in summer, or in Southern Hemisphere mid-winter) the effect is absent.

6. CONCLUSION

Efforts to predict future changes in the stratosphere are inevitably moving toward coupled chemistry-climate models (Austin et al. 2002). But dynamics remains a central issue. The patterns of variability and likely dynamical links between stratosphere and troposphere discussed in section 5 are part of increasing evidence that the stratosphere may play an active role in the tropospheric circulation, perhaps leading to changes in the troposphere as a result of stratospheric ozone changes (Hartmann et al. 2000) or stratospheric injections of volcanic aerosol (Robock 2000). These links may also offer scope for improving the skill of extended-range tropospheric forecasts (Baldwin et al. 2003).

A key outstanding dynamical problem in climate modelling for the stratosphere is reliable prediction of changes in the wave force \mathcal{G} . There is evidence from some numerical simulations with general circulation models that an increase in carbon dioxide implies an increase in planetary wave propagation into the stratosphere and hence an increase in the strength of the mean meridional circulation (e.g., Butchart & Scaife 2001, Rind et al. 2002 and references therein). However, this predicted increase is not certain—for example, Gillett et al. (2003) show an increase in wave propagation into the stratosphere only in the Northern Hemisphere, whereas Butchart & Scaife (2001) show changes in both hemispheres. Further (and possibly greater) potential uncertainty enters through the extreme difficulty in simulating possible changes in gravity wave sources in the troposphere.

ACKNOWLEDGMENTS

The author is particularly grateful to Michael McIntyre for detailed comments on a draft of this review. Other useful comments were received from Pavel Berloff, Steven Hardiman, and David Thompson.

The Annual Review of Fluid Mechanics is online at http://fluid.annualreviews.org

LITERATURE CITED

- Ambaum MHP, Hoskins BJ. 2002. The NAO troposphere-stratosphere connection. J. Climate 15:1969–78
- Andrews DG, Holton JR, Leovy CB. 1987. *Middle Atmosphere Dynamics*. Academic. 489 pp.
- Austin J, Shepherd TG, Schnadt C, Rozanov E, Pitari G, et al. 2003. Uncertainties and assessments of chemistry-climate models of the stratosphere. *Atmos. Chem. Phys.* 3:1–27
- Baldwin MP, Dunkerton TJ. 1999. Propagation of the Arctic Oscillation from the stratosphere to the troposphere. J. Geophys. Res. 104:30937–46
- Baldwin MP, Dunkerton TJ. 2001. Stratospheric harbingers of anomalous weather regimes. *Science* 294:581–84
- Baldwin MP, Gray LJ, Dunkerton TJ, Hamilton K, Haynes PH, et al. 2001. The quasi-biennial oscillation. *Rev. Geophys.* 39:179–229
- Baldwin MP, Stephenson DB, Thompson DWJ, Dunkerton TJ, Charlton AJ, O'Neill A. 2003. Stratospheric memory and skill of extendedrange weather forecasts. *Science* 301:636–40
- Black RX. 2002. Stratospheric forcing of surface climate in the Arctic Oscillation. J. Climate 15:268–77
- Boville BA. 1986. Wave-mean flow interactions in a general circulation model of the troposphere and stratosphere. J. Atmos. Sci. 43:1711–25
- Boville BA. 1995. Middle Atmosphere version of CCM2 (MACCM2)—annual cycle and interannual variability. J. Geophys. Res. 100:9017–39
- Bühler O, McIntyre ME. 2003. Remote recoil: a new wave-mean interaction effect. J. Fluid Mech. 492:207–30
- Butchart N, Austin J, Knight JR, Scaife AA, Gallani ML. 2000. The response of the stratospheric climate to projected changes in the concentrations of well-mixed greenhouse gases from 1992 to 2051. *J. Climate* 13:2142–59

- Butchart N, Scaife AA. 2001. Removal of chlorofluorocarbons by increased mass exchange between the stratosphere and troposphere in a changing climate. *Nature* 410:799– 802
- Chen P. 1996. The influences of zonal flow on wave breaking and tropical–extratropical interaction in the lower stratosphere. *J. Atmos. Sci.* 53:2379–92
- Christiansen B. 1999. Stratospheric vacillations in a general circulation model. *J. Atmos. Sci.* 56:1858–72
- Christiansen B. 2000. Chaos, quasiperiodicity, and interannual variability: studies of a stratospheric vacillation model. J. Atmos. Sci. 57:3161–73
- Dritschel DG. 1989. On the stabilization of a two-dimensional vortex strip by adverse shear. J. Fluid Mech. 206:193–221
- Dritschel DG, Haynes PH, Juckes MN, Shepherd TG. 1991. The stability of a twodimensional vorticity filament under uniform strain. J. Fluid Mech. 230:647–65
- Dritschel DG, Saravanan R. 1994. Threedimensional quasi-geostrophic contour dynamics, with an application to stratospheric vortex dynamics. Q. J. R. Meteorol. Soc. 120:1267–97
- Dunkerton TJ. 1989. Nonlinear Hadley circulation driven by asymmetric differential heating. J. Atmos. Sci 46:956–74
- Dunkerton TJ. 1991. Nonlinear propagation of zonal winds in an atmosphere with Newtonian cooling and equatorial wavedriving. J. Atmos. Sci. 48:236–63
- Dunkerton TJ, Hsu C-PF, McIntyre ME. 1981. Some Eulerian and Lagrangian diagnostics for a model stratospheric warming. *J. Atmos. Sci.* 38:819–43
- Feldstein S, Lee S. 1998. Is the atmospheric zonal index driven by an eddy feedback? J. Atmos. Sci. 55:3077–86
- Fritts DC, Alexander MJ. 2003. Gravitywave dynamics and effects in the middle

atmosphere. *Rev. Geophys.* 41:doi:10.1029/2001RG000106

- Garcia RR. 1987. On the mean meridional circulation of the middle atmosphere. J. Atmos. Sci. 44:3599–609
- Garcia RR, Boville BA. 1994. "Downward control" of the mean meridional circulation and temperature distribution of the polar winter stratosphere. J. Atmos. Sci. 51:2238–45
- Gillett NP, Allen MR, Williams KD. 2003. Modelling the atmospheric response to doubled CO₂ and depleted stratospheric ozone using a stratosphere-resolving coupled GCM. *Q. J. R. Meteorol. Soc.* 129:947–66
- Gray LJ. 2003. The influence of the equatorial upper stratosphere on stratospheric sudden warmings. *Geophys. Res. Lett.* 30:art. no. 1166
- Gray LJ, Sparrow S, Juckes M, O'Neill A, Andrews DG. 2003. Flow regimes in the winter stratosphere of the northern hemisphere. Q. J. R. Meteorol. Soc. 129:925–46
- Hartley DE, Villarin JT, Black RX, Davis CA. 1998. A new perspective on the dynamical link between the stratosphere and troposphere. *Nature* 391:471–74
- Hartmann DL, Wallace JM, Limpasuvan V, Thompson DWJ, Holton JR. 2000. Can ozone depletion and global warming interact to produce rapid climate change? *Proc. Natl. Acad. Sci. USA* 97:1412–17
- Haynes PH. 1990. High-resolution threedimensional modelling of stratospheric flows: quasi-two-dimensional turbulence dominated by a single vortex. In *Topological Fluid Mechanics*, ed. HK Moffatt, A Tsinober, pp. 345–54. Cambridge: Cambridge Univ. Press
- Haynes PH. 1998. The latitudinal structure of the quasi-biennial oscillation. Q. J. R. Meteorol. Soc. 124:2645–70
- Haynes PH, Marks CJ, McIntyre ME, Shepherd TG, Shine KP. 1991. On the "downward control" of extratropical diabatic circulations by eddy-induced mean zonal forces. *J. Atmos. Sci.* 48:651–78
- Haynes PH, Ward WE. 1993. The effect of realistic radiative transfer on potential vor-

ticity structures, including the influence of background shear and strain. J. Atmos. Sci. 50:3431–53

- Held IM, Hou AY. 1980. Nonlinear axially symmetric circulations in a nearly inviscid atmosphere. J. Atmos. Sci. 37:515–33
- Held IM, Phillips PJ. 1990. A barotropic model of the interaction between the Hadley cell and a Rossby wave. J. Atmos. Sci. 47:856– 69
- Holton JR, Mass C. 1976. Stratospheric vacillation cycles. J. Atmos. Sci. 33:2218–25
- Holton JR, Wehrbein M. 1980. The role of forced planetary waves in the annual cycle of the zonal mean circulation of the middle atmosphere. J. Atmos. Sci. 37:1968–83
- Holton JR, Wehrbein M. 1981. A further study of the annual cycle of the zonal mean circulation in the middle atmosphere. *J. Atmos. Sci.* 38:1504–9
- Holton JR, Haynes PH, McIntyre ME, Douglass AR, Rood RB, Pfister L. 1995. Stratosphere–troposphere exchange. *Rev. Geophys.* 33:403–39
- Juckes MN. 1989. A shallow water model of the winter stratosphere. J. Atmos. Sci. 46:2934– 55
- Juckes MN, McIntyre ME. 1987. A high resolution, one-layer model of breaking planetary waves in the stratosphere. *Nature* 328:590– 96
- Kodera K, Yamazaki K, Chiba K, Shibata K. 1990. Downward propagation of upper stratospheric mean zonal wind perturbation to the troposphere. *Geophys. Res. Lett.* 17:1263–66
- Legras B, Dritschel DG, Caillol P. 2001. The erosion of a distributed two-dimensional vortex in a background straining flow. *J. Fluid Mech.* 441:369–98
- Legras B, Joseph B, Lefévre F. 2003. Vertical diffusivity in the lower stratosphere from Lagrangian back-trajectory reconstructions of ozone profiles. J. Geophys. Res. 108:doi:10.1029/2002JD003045
- Limpasuvan V, Hartmann DL. 2000. Wavemaintained annular modes of climate variability. J. Climate 13:4414-2-9

- McIntyre ME. 1982. How well do we understand the dynamics of stratospheric warmings? J. Meteorol. Soc. Japan 60:37–65
- McIntyre ME. 2003a. Balanced flow. In *Encyclopedia of Atmospheric Sciences*, vol. 2, ed. JR Holton, JA Pyle, JA Curry. London: Academic/Elsevier
- McIntyre ME. 2003b. Potential vorticity. In *Encyclopedia of Atmospheric Sciences*, vol. 2, ed. JR Holton, JA Pyle, JA Curry. London: Academic/Elsevier
- McIntyre ME, Norton WA. 1990. Dissipative wave-mean interactions and the transport of vorticity or potential vorticity. J. Fluid Mech. 212:403–35 Corrigendum 220:693
- McIntyre ME, Palmer TN. 1983. Breaking planetary waves in the stratosphere. *Nature* 305:593–600
- McIntyre ME, Palmer TN. 1984. The "surf zone" in the stratosphere. J. Atmos. Terr. Phys. 46:825–49
- Newman PA, Rosenfield JE. 1997. Stratospheric thermal damping times. *Geophys. Res. Lett.* 24:433–36
- Norton WA. 2003. Sensitivity of northern hemisphere surface climate to simulation of the stratospheric polar vortex. *Geophys. Res. Lett.* 30: art. no. 1627
- O'Neill A, Pope VD. 1988. Simulations of linear and nonlinear disturbances in the stratosphere. Q. J. R. Meteorol. Soc. 114:1063–110
- Perlwitz J, Harnik N. 2003. Observational evidence of a stratospheric influence on the troposphere by planetary wave reflection. *J. Climate* 16:3011–26
- Plumb RA. 1977. The interaction of two internal waves with the mean flow: implications for the theory of the quasi-biennial oscillation. J. Atmos. Sci. 34:1847–58
- Plumb RA. 2002. Stratospheric transport. J. Meteorol. Soc. Japan 80:793–801
- Plumb RA, Eluszkiewicz J. 1999. The Brewer– Dobson circulation: dynamics of the tropical upwelling. J. Atmos. Sci. 56:868–90
- Plumb RA, Semeniuk K. 2003. Downward migration of extratropical zonal wind anomalies. J. Geophys. Res. doi: 10.1029/2002 JDOO2773

- Polvani LM, Dritschel DG. 1993. Wave and vortex dynamics on the surface of a sphere. *J. Fluid Mech.* 255:35–64
- Polvani LM, Plumb RA. 1992. Rossby wave breaking, microbreaking, filamentation and secondary vortex formation: the dynamics of a perturbed vortex. *J. Atmos. Sci.* 49:462– 76
- Polvani LM, Saravanan R. 2000. The threedimensional structure of breaking Rossby waves in the polar wintertime stratosphere. *J. Atmos. Sci.* 57:3663–85
- Polvani LM, Waugh DW, Plumb RA. 1995. On the subtropical edge of the stratospheric surf zone. J. Atmos. Sci. 52:1288–309
- Randel WJ. 1992. Global Atmospheric Circulation Statistics: 1000-1 mb. NCAR Tech. Note NCAR/TN-366+STR. Boulder, CO
- Randel WJ. 1993. Global variations of zonal mean ozone during stratospheric warming events. J. Atmos. Sci. 50:3308–21
- Randel WJ, Garcia RR, Wu F. 2002. Timedependent upwelling in the tropical lower stratosphere estimated from the zonal-mean momentum budget. *J. Atmos. Sci.* 59:2141– 52
- Rind D, Lonergan P, Balachandran NK, Shindell D. 2002. 2xCO₂ and solar variability influences on the troposphere through wavemean flow interactions. *J. Met. Soc. Japan* 80:863–76
- Robinson WA. 1991. The dynamics of the zonal index in a simple model of the atmosphere. *Tellus* 43A:295–305
- Robock A. 2000. Volcanic eruptions and climate. *Rev. Geophys.* 38:191–219
- Rosenlof KH. 1995. Seasonal cycle of the residual mean meridional circulation in the stratosphere. *J. Geophys. Res.* 100:5173–91
- Salby ML, Garcia RR, O'Sullivan D, Callaghan P. 1990. The interaction of horizontal eddy transport and thermal drive in the stratosphere. J. Atmos. Sci. 47:1647–65
- Scaife AA, James IN. 2000. Response of the stratosphere to interannual variability of tropospheric planetary waves. Q. J. Roy. Meteorol. Soc. 126:275–97

- Scinocca JF, Haynes PH. 1998. Dynamical forcing of planetary waves by tropospheric baroclinic eddies. J. Atmos. Sci. 55:2361–92
- Scott RK. 2002. Wave-driven mean tropical upwelling in the lower stratosphere. J. Atmos. Sci. 59:2745–59
- Scott RK, Haynes PH. 1998. Internal interannual variability of the extratropical stratospheric circulation: the low-latitude flywheel. Q. J. R. Meteorol. Soc. 124:2149– 73
- Scott RK, Haynes PH. 2000. Internal vacillations in stratosphere-only models. J. Atmos. Sci. 57:2333–50
- Scott RK, Haynes PH. 2002. The seasonal cycle of planetary waves in the winter stratosphere. J. Atmos. Sci. 59:803–22
- Scott RK, Polvani LM. 2004. Stratospheric control of upward wave flux near the tropopause. *Geophys. Res. Lett.* 31:doi:10.1029/2003 GL017965
- Semeniuk K, Shepherd TG. 2001. The middleatmosphere Hadley circulation and equatorial inertial adjustment. J. Atmos. Sci. 58: 3077–96
- Semeniuk K, Shepherd TG. 2001. Mechanisms for tropical upwelling in the stratosphere. J. Atmos. Sci. 58:3097–115
- Simmons A, Hortal M, Kelly G, McNally A, Untch A, Uppala S. 2004. ECMWF analyses and forecasts of stratospheric winter polar vortex break-up: September 2002 in the Southern Hemisphere and related events. *J. Atmos. Sci.* 61:In press
- Song Y, Robinson WA. 2004. Dynamical mechanisms of stratospheric influences on the troposphere. J. Atmos. Sci. 61:1711–25
- Taguchi M, Yamaga T, Yoden S. 2001. Internal variability of the troposphere-stratosphere coupled system simulated in a simple global circulation model. *J. Atmos. Sci.* 58:3184– 203
- Taguchi M, Yoden S. 2002. Internal interannual variability of the troposhere-stratosphere coupled system in a single global circulation model. Part I: parameter sweep experiment. *J. Atmos. Sci.* 59:3021–36

Taguchi M, Yoden S. 2002. Internal interannual

variability of the troposhere-stratosphere coupled system in a simple global circulation model. Part II: millennium integrations. *J. Atmos. Sci.* 59:3037–50

- Thompson DWJ, Wallace JM. 2000. Annular modes in the extratropical circulation. Part I: month-to-month variability. *J. Climate* 13:1000–16
- Thuburn J, Lagneau V. 1999. Eulerian mean, contour integral, and finite-amplitude wave activity diagnostics applied to a single-layer model of the winter stratosphere. *J. Atmos. Sci.* 56:689–710
- Tung KK, Kinnersley JS, 2001. Mechanisms by which extratropical wave forcing in the winter stratosphere induces upwelling in the summer hemisphere. *J. Geophys. Res.* 106:22781–92
- Waugh DW, Dritschel DG. 1999. The dependence of Rossby wave breaking on the vertical structure of the polar vortex. J. Atmos. Sci. 56:2359–75
- Waugh DW, Randel WJ, Pawson S, Newman PA, Nash ER. 1999. Persistence of the lower stratospheric polar vortices. J. Geophys. Res. 104:27191–201
- WMO (World Meteorological Organisation). 1999. Scientific assessment of ozone depletion 1998. WMO Global Ozone Res. Monit. Proj. Rep. 44. World Meteorol. Org., Geneva
- WMO (World Meteorological Organisation). 2003. Scientific assessment of ozone depletion 2002. WMO Global Ozone Res. Monit. Proj. Rep. 47. World Meteorol. Org., Geneva
- Yoden S. 1987. Bifurcation properties of a stratospheric vacillation model. *J. Atmos. Sci.* 44:1723–33
- Yoden S. 1990. An illustrative model of seasonal and interannual variations of the stratospheric circulation. J. Atmos. Sci. 47:1845– 53
- Yoden S, Taguchi M, Naito Y. 2002. Numerical studies on time variations of the tropospherestratosphere coupled system. J. Met. Soc. Japan 80:811–30
- Yulaeva E, Holton JR, Wallace JM. 1994. On the cause of the annual cycle in tropical

lower stratospheric temperatures. J. Atmos. Sci. 51:169–74

Zeng G, Pyle JA. 2003. Changes in tropospheric ozone between 2000 and 2100 modelled in a chemistry-climate model. *Geophys. Res. Lett.* 30: art. no. 1392

Zhu X. 1993. Radiative damping revisited-

parametrization of damping rate in the middle atmosphere. J. Atmos. Sci. 50:3008– 21

Zhu X. 1997. Comments on "A New Parameterization of Scale-Dependent Radiative Rates in the Stratosphere." *J. Atmos. Sci.* 54:1388– 92